Author’s response to reviews

Title: Comparing databases: determinants of sexually transmitted infections, HIV diagnoses, and lack of HIV testing among men who have sex with men

Authors:

Chantal C den Daas (chantal.den.daas@rivm.nl)
Maaike M Goenee (m.goenee@rutgerswpf.nl)
Bouko BHW Bakker (b.bakker@live.nl)
Hanneke H de Graaf (h.degraaf@rutgerswpf.nl)
Eline ELM Op de Coul (eline.op.de.coul@rivm.nl)

Version: 5
Date: 19 October 2015

Author’s response to reviews: see over
Dear Dr. Minichiello,

We would like to thank you for your reaction regarding our manuscript (MS# 2501535031671181), entitled “Comparing databases: determinants of sexually transmitted infections, HIV diagnoses, and lack of HIV testing among men who have sex with men”. We are happy to submit a revised version of the manuscript. We have added the requested reference (line number 152, reference number 17). This is a reference to the Dutch site that exactly explains when research should apply for ethical approval and when not (in Dutch “uw onderzoek wmo plichtig of niet”).

We hope that you can consider the revised version of the manuscript for publication in BMC Public Health.

Sincerely, also on behalf of my co-authors,

Chantal den Daas